Author’s response to reviews

Title: Cognitive function, social integration and mortality in a U.S. national cohort study of older adults

Authors:

Thomas O. Obisesan (tobisesan@howard.edu)
R. F. Gillum (frank.gillum@gmail.com)

Version: 4 Date: 17 July 2009

Author’s response to reviews:

Dear Editors:

We thank the Editors and reviews for their favorable consideration of our submission for publication in BMC Geriatrics as a Research Article entitled, “Cognitive Function, Social Integration and Mortality in a U.S. National Cohort Study of Older Adults.” We have made the requested formatting changes. We would appreciate the journal removing vertical lines from the tables and cropping the figures as needed. We look forward to publication.

Please address all correspondence to me at my email address.

Sincerely,

R. Frank Gillum, MD, MS